Author’s response to reviews

Title: Overexpression of members of the microRNA-183 family is a risk factor for lung cancer: a case control study

Authors:

Zhu Wangyu (zhuwangyu24@gmail.com)
Liu Xiaoguang (lxg2004@gmail.com)
He Jianying (hiyncmcstudent@163.com)
Chen Dongdong (wydxzhishi@sina.cn)
Huang Yanyan (huangyanyan2005@sina.cn)
Zhang Yongkui (zyk801801@126.com)

Version: 3 Date: 30 August 2011

Author’s response to reviews: see over
Dear Editor Danrolf,

Thank you very much for your advice. We wrote the name again according to the right format of the manuscript. Otherwise, we added the sentence "The authors would like to thank Medjaden Bioscience Limited for assisting in the preparation of this manuscript" in the acknowledgements section and removed all highlighting and tracking from the manuscript. Lastly, we also checked the typographical errors and formats of the manuscript. We hope the paper will be published early.

Thank you very much.

With best wishes,

Yours sincerely,

Wangyu Zhu

Corresponding author: Yongkui Zhang

Department of Cardio-Thoracic Surgery; Zhoushan Hospital of Zhejiang Province, Zhoushan, Zhejiang, 316004, China;

Fax: +86-580-2558005; Phone: +86-580-2558362

E-mail: zyk801801@126.com, zhuwangyu24@gmail.com